Author’s response to reviews

Title: Inferior Alveolar Nerve Injury With Laryngeal Mask Airway-A Case Report

Authors:

Deepak Hanumanthaiah (drdeepakhrh@yahoo.co.in)
Sarmad Masud (docsarmad@hotmail.com)
anil ranganath (dranil_2@rediffmail.com)

Version: 4 Date: 24 October 2010

Author’s response to reviews:

1] An introduction about why this case is important and needs to be reported. Please include information on whether this is the first report of this kind in the literature: This change is made. The fact that this is the first case of alvelolar nerve damage with LMA is mentioned in introduction. Changes are highlighted.

2] Case presentation: the age and ethnicity of the patient is mentioned in the abstract. Changes are highlighted.

3] Conclusion: the information that it is the first article of its kind and of interest particularly for anaesthetist is mentioned. All the changes are highlighted.